Author's response to reviews

Title: Development of the Clinical Learning Evaluation Questionnaire for undergraduate clinical education: factor structure, validity, and reliability study.

Authors:

Ali I AlHaqwi (alhaqwi@gmail.com)
Jeroen Kuntze (kuntze@fsw.eur.nl)
Henk Van der Molen (vandermolen@fsw.eur.nl)

Version: 2 Date: 1 June 2013

Author's response to reviews:

Please note the following changes:

1- Title was modified.

2- Institutes of authors and corresponding author were mentioned in the title page.

3- Statement about the approval of ethics' committee was added in the method section.

4- Statement about competing interest was added.

5- Authors contributions were clarified.

6. Acknowledgement section was added.